Author's response to reviews

Title: Aqueous carob pods extract protects against ethanol-induced gastric ulcer in rat Kais Rtibi a,b, Abdelaziz Souli a,b, Hichem Sebai c, Mohamed Amine Jabri c, Slimen Selmi b, Jamel El-Benna d, Mohamed Amri a and Mohamed Lamjid Marzouki a,b

a Laboratoire de Neurophysiologie Fonctionnelle et Pathologies, Departement des Sciences Biologiques, Faculte des Sciences de Tunis. Campus Universitaire El Manar II-2092 Tunis, Tunisia. b Laboratoire de Nutrition et Physiologie Animale, Institut Superieur de Biotechnologie de Beja, Avenue Habib Bourguiba, B.P., 382-9000 Beja, Tunisia. c Laboratoire de Physiologie Integree, Faculte des Sciences de Bizerte, 7021 Zarzouna, Tunisia. d INSERM U773 Centre de Recherche Biomedicale, Faculte de Medecine X. Bichat, 75018 Paris, France. *Corresponding author at: Dr. Kais Rtibi, Laboratoire de Neurophysiologie Fonctionnelle et Pathologies, Departement des Sciences Biologiques, Faculte des Sciences de Tunis. Campus Universitaire El Manar II-2092 Tun

Authors:

Kais Rtibi (rtibikais@yahoo.fr)
Abdelaziz Souli (AbdelazizSouli@fsg.rnu.tn)
Hichem Sebai (sebai.Hichem@yahoo.fr)
Mohamed Amine Jabri (jabriamino@gmail.com)
Slimen Selmi (slimen.selmi@gmail.com)
Jamel El-Benna (jamelelbenna@inserm.fr)
Mohamed Amri (mohamedamri@fst.rnu.tn)
Mohamed Lamjid Marzouki (lamjed.marzouki@ipeis.rnu.tn)

Version: 3
Date: 11 February 2015

Author's response to reviews:

1. The scientific justification for the analysis of gastroprotective effects of carob pods in rats was confirmed.

2. The animals were used in these experiments in accordance with the ethic committee of Tunis University for care and use of animals in conformity with NIH guideline.

3. The title of conclusion is included.

4. The Voucher specimens have been deposited with the herbarium of the Higher Institute of Biotechnology of Beja and also in Department of Biological Sciences, Faculty of Science, Tunisia.

5. The title page at the front of the manuscript file contains at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.